

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Cici, Gjergji; Shane, Philip B.; Yang, Yanhua Sunny

Working Paper Do connections with buy-side analysts inform sell-side analyst research?

CFR Working Paper, No. 17-04

Provided in Cooperation with: Centre for Financial Research (CFR), University of Cologne

Suggested Citation: Cici, Gjergji; Shane, Philip B.; Yang, Yanhua Sunny (2017) : Do connections with buy-side analysts inform sell-side analyst research?, CFR Working Paper, No. 17-04, University of Cologne, Centre for Financial Research (CFR), Cologne

This Version is available at: https://hdl.handle.net/10419/171277

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Do Connections with Buy-Side Analysts Inform Sell-Side Analyst Research?

Gjergji Cici*

Thomas L. Owen Associate Professor of Finance College of William and Mary Raymond A. Mason School of Business Gjergji.Cici@mason.wm.edu

Philip B. Shane KPMG Professor of Accounting College of William and Mary Raymond A. Mason School of Business Phil.Shane@mason.wm.edu

Yanhua Sunny Yang

Associate Professor of Accounting University of Connecticut School of Business Yanhua.yang@uconn.edu

Abstract: We hypothesize that connections with buy-side analysts provide a sell-side analyst with private information generated by the buy-side that enhances the quality of sell-side research. We proxy for these connections with the number of stocks at the intersection of stocks held in the portfolios of institutional investors and followed by the sell-side analyst. The larger this intersection, the more opportunities the sell-side analyst has to interact with institutional investors. We proxy for the research quality of the sell-side analyst with the relative accuracy of her earnings forecasts. We find that such connections enhance the accuracy of earnings forecasts, but up to a point of diminishing returns. Additional tests rule out that the observed association is due to reverse causality.

First Draft: 3 August 2017 This Draft: 11 October 2017

^{*} Cici is also a research fellow at the Centre for Financial Research at the University of Cologne. We appreciate comments from participants at workshops at Monash University and the University of Queensland. Shane gratefully acknowledges support from the Frank Wood Accounting Research Fund at the College of William and Mary.

Do Connections with Buy-Side Analysts Inform Sell-Side Analyst Research?

1. Introduction

This paper investigates whether, through connections with buy-side analysts, sell-side analysts glean information that enhances the quality of their research reports. A vast literature studies the characteristics of the research produced by sell-side analysts and its impact on stock prices, arguably through its impact on institutional investor decisions.¹ This research generally assumes—and indeed some studies show— that the flow of information between sell-side and buy-side analysts is one-directional; i.e., information supplied by the sell-side flows to institutional investors, whose trades move stock prices (e.g., Gu, Li, Li, & Yang 2016; Irvine, Lipson, & Puckett, 2007; Mikhail, Walther, & Willis, 2007). Our paper looks at the flow of information in the other direction; i.e., do insights from the research of buy-side analysts, in support of institutional investor decisions, flow to sell-side analysts and improve the quality of outputs from sell-side analyst research reports?

We hypothesize that interactions with buy-side analysts provide sell-side analysts with private information generated by the buy-side that enhances the quality of sell-side research. Although buy-side analysts' research is proprietary and therefore not publicly available to other market participants, there are two factors that make sell-side analysts privy to at least part of this information.

First, learning what other buy-side analysts think and sharing that with institutional clients is an implicit service expected of sell-side analysts. Brown, Call, Clement, & Sharp (2016) surveyed and interviewed buy-side analysts who indicated that their demand for sell-side

¹ Recent papers of this ilk include Bradley, Gokkaya, & Liu (2017), Merkley, Michaely, & Pacelli (2017), Horton, Serafeim, & Wu (2017), Bernhardt, Wan, & Zhao (2016), and Franck & Kerl (2013). Earlier papers providing thorough reviews of research with that perspective include Ramnath, Rock, & Shane (2008a, 2008b) and Bradshaw (2011).

analyst services depends, primarily, on: (i) the ability of sell-side analysts to facilitate meaningful one-on-one interaction with CFOs and other knowledgeable executives working for the firms whose securities represent important components of fund manager portfolios;² (ii) the quality of sell-side analysts' industry-related research; and (iii) insights sell-side analysts provide into the perspective of buy-side analysts working for other institutional investment firms. Institutional investors could, and increasingly do, internalize the first two services, but they must outsource the third service, which incentivizes sell-side analysts to discover their buy-side analyst clients' perspectives on the firms the sell-side analysts follow.

Second, sell-side analysts have many opportunities to learn from buy-side analysts. The lion share of a typical sell-side analyst's compensation is driven by broker votes, and broker votes are in turn driven by personalized services that sell-side analysts provide for institutional clients such as high-touch meetings, phone calls, whitepapers, and concierge services that put buy-side analysts in touch with management of the firms of interest to them (Maber, Groysberg, & Healy 2015). Thus, analysts have a strong incentive to meet their institutional clients' needs by providing high-touch services, which necessitate regular communication with current or potential institutional investor clients. These communications provide a window through which sell-side analysts can infer buy-side private information about covered firms. For example, some clients might follow up with a sell-side analyst to ask for more color following her recently issued research report, updated earnings' estimates, or change in a stock recommendation, while other clients might want to ask questions related to company-specific or industry-specific

² A recent survey indicates that one-third of the portion of commission payments used to compensate brokerage firms rewards sell-side analysts for corporate access, including facilitation of meetings between buy-side analysts and company management ... (Greenwich Associates, 2010). Another survey finds that CEOs, CFOs, and investor relation professionals consider facilitating meetings with the buy-side as the most critical aspect of the investor relations function (Thomson Reuters 2009 IR Best Practices).

developments, all of which provide a sell-side analyst with opportunities to uncover and put together various pieces of information produced by various institutional investors.

As rational economic agents, sell-side analysts have strong incentives to use the private information they acquire from their interactions with buy-side analysts to their advantage. Information gleaned from communications with buy-side analysts can provide important input to sell-side analyst research and, ultimately, enhance the quality of the sell-side analyst research report. In this regard, Groysberg, Healy, & Chapman (2008) speculate that "sell-side analysts may develop an information advantage through feedback on their ideas from their own institutional clients (p. 33)." By employing the private information gleaned from buy-side analysts and using it to enhance the quality of their whitepapers and research reports, sell-side analysts can do a better job articulating a deep understanding of the value-relevant affairs of the firms s/he covers.³ This in turn would generate more interest among institutional clients and consequently more brokerage votes and higher compensation for the sell-side analysts.

The most salient component of the sell-side analyst research report is the analyst's forecast of the followed firm's earnings for the remaining quarters of the current fiscal year. We use these earnings forecasts to proxy for the quality of the sell-side analyst's research report, and their accuracy determines the dependent variable in our study.⁴ We measure accuracy at the analyst-firm-year level and refer to this dependent variable as *ACCURACY*. In measuring this

³According to Maber, et al. (2015), "whitepapers, despite being non-timely and vulnerable to significant publicgoods problems, are an important vehicle through which analysts signal their industry knowledge and build and sustain their franchise." Maber et al. find that publication of whitepapers explains more variation in broker votes than any other aspect of sell-side analyst output.

⁴ Earnings forecasts represent the most prevalent form of research output, and information in earnings forecasts affects the quality of other research outputs: i.e., analysts' stock recommendations (e.g.,Ertimur, Sunder, & Sunder 2007) and target price forecasts (e.g., Gleason, Johnson, and Li 2013). Thus, we believe earnings forecast accuracy is a reasonable proxy for the overall quality of the sell-side analyst research report. Second, evaluating the ability to forecast earnings is straightforward since the forecast is always compared against a deterministic benchmark, i.e., actual earnings. The same cannot be said for stock recommendations where one has to choose a performance evaluation model and horizon, or for the target price forecasts where the benchmark, the actual price at the end of the 12-month period, might not correctly reflect the fair value of the stock due to volatile market conditions.

variable, we hold constant the firm and year and measure analyst a's accuracy relative to all other analysts following firm f's stock in year t. Our multivariate regressions control for other factors known from prior literature to affect analyst earnings forecast accuracy.

Key to our identification approach is an independent variable, which measures the strength of holdings-based connections between sell-side analysts and institutional investors. Given that higher institutional ownership gives rise to more accurate earnings forecasts by sellside analysts due to a higher demand for more accurate research (see e.g., Frankel, Kothari, and Weber, 2006; Ljungqvist, Marston, Starks, Wei, and Yan (2007); and Wong (2016)), we construct our connections measure to bypass the confounding effect of institutional ownership on forecast accuracy. Specifically, for each stock, f, covered by sell-side analyst a, our measure of connections does not rely on the number of institutions that hold f. Instead, if f is held in the portfolio of institutional investor, *i* during year *t*, the number of other stocks followed by *a* and held by *i* is our proxy for the strength of the connection between *a* and *i* during year t.⁵ The average of this number, across all institutions holding f, is our key independent variable, CONNECTIONS_{aff}. We expect that the more connections an analyst has with institutional investors, the greater the likelihood of communications with these investors and the greater the opportunity for the sell-side analyst to decipher the private information possessed by institutional investors. A significantly positive relation between ACCURACYaft and CONNECTIONSaft supports the hypothesis that sell-side analysts learn from their interactions with their buy-side clients and that learning process enhances the quality of the sell-side analyst's research reports.

As explained in Section 2 below, we hypothesize and find that the relation between $ACCURACY_{aft}$ and $CONNECTIONS_{aft}$ weakens as $CONNECTIONS_{aft}$ reaches a point of

⁵ Throughout this paper we refer to the issuing firm and the firm's stock with the same subscript, f. We use the subscript, i, to refer to the institutional investors that employ buy-side analysts. The subscript, a, refers to the sell-side analyst, the subscript, t, refers to the year, and our analysis is based on analyst-firm-years, *aft*.

diminishing returns. This result is analogous to prior research that finds lower levels of earnings forecast accuracy among sell-side analysts who cover large numbers of firms. It is also consistent with Maber, Groysberg, & Healy 2015, who show that increased interactions with institutional clients associated with high-touch services come with opportunity costs from not being able to spend time on other aspects of their sell-side analyst jobs.

From evidence consistent with the hypothesized non-linear relation between *ACCURACY* and *CONNECTIONS*, we infer that information from connections with buy-side analysts informs sell-side analyst research. However, the discussion above suggests an endogenous relation between *ACCURACY* and *CONNECTIONS*, whereby buy-side analysts select sell-side analysts who can provide insights that inform the buy-side analysts' research reports to their fund managers, and that selection probably favors sell-side analysts who have already proven themselves in ways that might include forecast accuracy. To support our inferences, we need to address this endogeneity issue.

We address the endogeneity issue in four ways. First, we examine whether the strength of the relation between *ACCURACY* and *CONNECTIONS* depends on the amount of private information that buy-side analysts possess. A stronger positive relation between *ACCURACY* and *CONNECTIONS* for buy-side analysts with more private information validates the inference that sell-side analysts obtain forecast accuracy-enhancing information from buy-side analysts. On the other hand, stronger relation with *CONNECTIONS* for buy-side analysts with less private information would support the inference that buy-side analyst selection of already-informed and accurate sell-side analysts drives the effect we document. Our proxy for the degree to which buy-side analysts possess private information measures the degree to which institutional investors rely on private versus public information to support their trading decisions (Kacperczyk & Seru,

2007). Our results support our hypotheses that the relation between *ACCURACY* and *CONNECTIONS* strengthens with increased reliance on private information by connected institutional investors.

Second, we examine the implications of an exogenous shock hypothesized to increase sell-side analyst demand for information from buy-side analysts. The shock we exploit appears during 2001-2003 when: (i) Regulation Fair Disclosure (Reg FD) came into effect constraining both sell-side and buy-side analyst ability to obtain private information from management of the firms they follow; and (ii) the Global Analyst Research Settlement (GARS) further constrained sell-side analysts' ability to obtain private information from investment bankers in their own firms. If buy-side selection of accurate sell-side analysts drives the relation between ACCURACY and CONNECTIONS, we see no reason for the relation to change in the wake of the 2001-2003 regulatory period. In fact, the regulation potentially eliminates sell-side ability to provide buyside analysts with access to private information held by management of the covered firm, and this is a key reason for buy-side interest in connecting with sell-side analysts (Brown et al. 2016). In addition, GARS eliminates communications between sell-side (not buy-side) analysts and investment bankers within their own firms. This increases the relative importance of buy-side analysts as a source of private information for sell-side analysts. Consistent with our main hypothesis that connections with buy-side analysts enhance sell-side analyst earnings forecast accuracy, we find an insignificant relation between ACCURACY and CONNECTIONS in the pre-2001 period and a significantly positive relation in the post-2003 period when sell-side analysts have more reason to seek private information from buy-side analysts.

Third, we examine factors hypothesized to increase buy-side analyst demand for connections with sell-side analysts (i.e., past sell-side forecast accuracy and sell-side analyst

prior experience following the covered firm), and we find no evidence to support the inference that increased buy-side analyst demand strengthens the relation between *ACCURACY* and *CONNECTIONS* documented in our primary hypothesis tests.

Finally, in additional analysis, we find that our main finding is robust in an analysis that restricts the sample to analysts with less than four years of firm-specific experience. In those cases, the buy-side analyst has very little basis for judging the accuracy track record of the sell-side analyst with whom s/he chooses to work. Also in additional analyses our inferences remain unchanged when we use two alternative proxies for our *CONNECTIONS* variable, thus strengthening our confidence in the construct validity of our measure of connectedness.

This paper contributes to the literature in two important ways. First, we contribute to the stream of research investigating the factors affecting sell-side analyst earnings forecast accuracy. Prior research essentially identifies three factors: (i) incentives (e.g., Michail, Walther, & Willis 1999), (ii) ability (e.g., Jacob, Lys, & Neale 1999; Clement 1999), and (iii) resources (e.g., Jacob, Rock, & Weber 2008). We introduce a fourth factor; i.e., accuracy-enhancing information obtained through interactions with buy-side analysts. Second, and more importantly, we open the door to a new avenue of research that can investigate the role of the bilateral flow of information between sell- and buy-side analysts in increasing the quality of information impounded in capital asset prices.

The rest of this paper is organized as follows. The next section reviews the literature and presents our hypotheses. We discuss our research design in Section 3 and in Section 4 present our sample selection procedure. Section 5 discusses the results of our hypotheses tests. Section 6 presents additional analyses, and Section 7 concludes.

2. Hypotheses and Literature Review

Main hypothesis

The Brown et al. (2016) survey suggests that buy-side analysts rely on sell-side analysts, primarily, for industry-related information, access to the management of the firms the sell-side analysts cover, and insight into the perspectives of buy-side analysts working for competing institutional investor firms. Thus, it appears that what buy-side analysts gain from their interactions with sell-side analysts informs a thorough buy-side analyst research process that finds little use for the stock recommendations produced by sell-side analysts. In spite of the fact that they appear to rely primarily on their own research, buy-side analysts regularly interact with sell-side analysts. When Brown et al. asked buy-side analysts how often they have private communication with sell-side analysts, they found that only 4% of their respondents said "never," and 55% said more than 23 times per year. Maber, Groysberg, & Healy 2015 analysis of sell-side analysts at a mid-size investment banking firm suggest that these interactions could be even more substantial when viewed from the perspective of the sell-side analysts: the average sell-side analyst holds approximately 750 private calls and 45 one-on-one meetings with client investors in the course of a typical semiannual period. We expect that, in the course of these interactions, sell-side analysts glean information useful for their own stock recommendations and earnings forecasts.

We argue that the regular communications with their institutional clients provide sell-side analysts with a window into the private information generated by their institutional clients about companies of common interest, which in turn improves the quality of sell-side research. Given the results of the Brown et al. (2016) survey indicating that buy-side analysts have little, if any, interest in the stock recommendations of sell-side analysts, we expect institutional investors to

have information that does not overlap with information sell-side analysts possess. Bushee, et al. (2016) document that trade sizes around investor-management meeting times increase and abnormal net buys around the meetings are profitable during thirty days subsequent to the private access day. They conclude that the private access to management provides information that changes institutional investors' beliefs and their trading. Such beliefs-changing information, which is unlikely to be in the information set of sell-side analysts could be "mosaic" but, nonetheless, valuable in combination with investors' private information and does not violate "Reg FD" (Solomon & Soltes, 2015).

Sell-side analysts could acquire information from their institutional clients informally. Specifically, as both parties engage in conversations, the questions raised and the requests for clarifications made by the institutional clients tip off sell-side analysts about the private information of their institutional clients. That sell-side analysts discern the private information of their institutional clients in the course of such communications is supported by the fact that many buy-side analysts view the knowledge that sell-side analysts have of other buy-side analysts' opinions as a valuable service provided by the sell-side (Brown et al. 2016). Furthermore, the results of the Brown et al. interviews suggest that buy-side analysts value their relationships with sell-side analysts, because "they are the only portal" into the thinking of buy-side analysts working for other institutions. Quoting one of their interviewees, "The buy side is this whole poker game of, 'I don't want to show my cards, but I want to see your cards.' The only people that can actually see everyone's cards is the sell side. When we ask them questions, they can figure out what we're thinking."

As described in Section 1, we develop a novel approach to measure the quantity of private information that sell-side analysts acquire from the interactions with their current or

potential buy-side clients. Our measure is specific to each sell-side analyst, a, and each stock, f, that analyst a follows in year t. For each stock, f, followed by sell-side analyst, a, and held in the portfolio of institutional investor, i during year t, the number of other stocks followed by a and held by i is our proxy for the strength of the connection between a and i during year t. The average of this number, across all institutions holding f, is our key independent variable, *CONNECTIONS*_{aff}. The premise of our analysis is that the more connections an analyst has with institutional investors, the greater the likelihood of communications with these investors and the greater the opportunity for the sell-side analyst to learn and decipher the private information possessed by institutional investors.

While sell-side analysts potentially learn more by talking more with buy-side analysts, it's costly for analysts to spread themselves too thinly. For example, there appears to be a cost associated with following too many firms (Clement 1999; Jacob, Lys, & Neale 1999; Myring & Wrege 2011; Pelletier 2015). We expect that for each sell-side analyst there is an optimal allocation of her efforts between conducting research and providing services for her institutional clients. Beyond that point, extra connections with buy-side analysts are likely to come with an opportunity cost that outweighs the benefit of other sell-side analyst who matter most, connecting with management of the firms they follow, and writing whitepapers and research reports. This is consistent with Maber, et al. (2015) who show that increases in analysts' time-consuming services for their institutional clients result in less published research output. Thus, we expect each sell-side analyst to have an optimal number of buy-side analysts from which s/he can glean incremental private information about any given stock. Beyond that number, we expect

diminishing returns as the number of interactions with different buy-side analysts increases. In light of this reasoning, we hypothesize the following relation:

H1: $ACCURACY_{aft}$ increases with $CONNECTIONS_{aft}$ up to some point where the rate of increase subsides.

Additional hypotheses

Our *CONNECTIONS* variable proxies for the flow of useful information from buy-side to sell-side analysts, where more information increases the quality of the sell-side analyst research proxied by greater forecast accuracy. Given evidence of the relation hypothesized in H1, we test additional hypotheses that identify factors expected to strengthen the relation between *ACCURACY* to *CONNECTIONS*. We develop these additional hypotheses to provide more confidence in the validity of the relation we observe in tests of H1 and to address the endogeneity issue discussed in Section 1. That is, the tests are designed to sort out whether the relation observed in tests of H1 emerges from sell-side analysts obtaining accuracy-enhancing information from buy-side analysts, or from buy-side analysts seeking connections with already-accurate sell-side analysts.

We predict greater sensitivity of *ACCURACY* to *CONNECTIONS* in situations where sell-side analysts have *more* opportunities to learn from their buy-side analyst counterparts, which would arise when sell-side analysts are connected with certain buy-side analysts who produce relatively *large* amounts of private information. If, on the other hand, *ACCURACY* drives *CONNECTIONS* because buy-side analysts have more need for information from sell-side analysts, then we expect greater sensitivity of *ACCURACY* to *CONNECTIONS* in situations where sell-side analysts have *less* opportunities to learn from their buy-side analyst counterparts. Such situations arise when sell-side analysts are connected with certain buy-side analysts who

produce relatively *small* amounts of private information. This discussion leads to our second hypothesis:

H2: The sensitivity of $ACCURACY_{aft}$ to $CONNECTIONS_{aft}$ increases with the opportunity for sell-side analysts to learn from buy-side analysts.

To identify institutional investors that produce more versus less private information, we draw on previous research. In particular, we rely on Kacperczyk & Seru (2007), who document that institutions less reliant on public information produce more private information, which results in superior investment performance relative to institutions that rely more on public information.

Since October 2000, Reg. FD has prohibited transmission of private information to sellside analysts from management of the firms they follow, and since April 2003, the Global Analyst Research Settlement (GARS) between the SEC and the 10 largest investment banks, presumably employing the largest number of sell-side analysts, has prohibited sell-side analysts from discussing the prospects of the firms they cover with the investment bankers within their own firms. We expect that these exogenous shocks considerably strengthened sell-side analyst incentives to obtain private information through discussions with buy-side analysts. These incentives emerged in a new environment where on one hand sell-side analysts were cut off from private information previously obtained from management of the firms they followed and investment bankers within their own firms, and on the other hand sell-side analysts came under increased pressure to generate commission revenue for their firms. If the relation we observe in tests of H1 above emanates from buy-side demand for connections with already-accurate sellside analysts, then we do not expect the relation to strengthen during the years after GARS. Neither Reg FD nor GARS affected buy-side analyst discussions with fund managers within their own firms, but these regulations did affect discussions between sell-side analysts and

investment bankers within their firms. Furthermore, Reg FD removed one of the buy side's benefits from interacting with sell-side analysts; i.e., obtaining firm-specific private information from the firm's management. This leads us to our third hypothesis:

H3: The sensitivity of *ACCURACY* to *CONNECTIONS* increases in the wake of Reg FD and GARS.

On the other hand, if buy-side analysts' demand for information from already-accurate sell-side analysts drives the relation between *ACCURACY* and *CONNECTIONS*, then we expect the relation to vary with factors hypothesized to reflect future forecast accuracy. This leads us to our fourth hypothesis:

H4: The sensitivity of *ACCURACY* to *CONNECTIONS* increases with increases in buyside analyst demand for connections with sell-side analysts.

We test H4 with reference to two factors hypothesized to predict sell-side analyst forecast accuracy, which in turn could drive buy-side analyst demand for connections. Prior research finds strong relations between sell-side analyst forecast accuracy and: (i) past accuracy (Brown 2001); and (2) firm-specific experience (Clement 1999). In response to the Brown, et al. (2016) survey, buy-side analysts rate the sell-side analyst's firm-specific experience as the most important attribute affecting the decision to use information provided by the sell-side analyst. In fact, this attribute is rated as more important than how often the sell-side analyst speaks with firm management, and whether the sell-side analyst is a member of the Institutional Investor All-American Research Team. Thus, if buy-side demand drives the relation between *ACCURACY* and *CONNECTIONS*, then we expect to find evidence supporting H4; i.e., we expect the relation to strengthen with sell-side analyst past accuracy and firm-specific experience.

3. Research design

3.1 Models for testing H1

If analysts produce more accurate forecasts due to the private information they collect from their connections with institutional investors, we expect $\beta_1 > 0$ in model (1) below. In addition, if analysts face diminished returns beyond a certain optimal level of connections with institutional investors, we expect $\beta_2 < 0$.

$$ACCURACY_{aft} = \beta_0 + \beta_1 CONNECTIONS_{aft} + \beta_2 CONNECTIONS_{aft}^2 + \sum_m \beta_m Control_m + \varepsilon_{aft}$$
(1)

Where:

$$ACCURACY_{aft} = \frac{\max(|FE_{ft}|) - |FE_{aft}|}{\max(|FE_{ft}|) - \min(|FE_{ft}|)}$$

 $\max(|FE_{ft}|)$ = the maximum absolute forecast error among all analysts issuing forecasts of firm *f*'s year *t* earnings during the first 90 days following the firm's year *t*-1 earnings announcement.⁶

 $min(/FE_{ft}/)$ = the minimum absolute forecast error from the distribution generating $max(/FE_{ft}/)$.

 $FE_{aft} = F_{aft} - A_{ft}$ = the error in analyst *a*'s most recent I/B/E/S-provided forecast, F_{aft} , of firm *f*'s year *t* earnings during the first 90 days of the firm's fiscal year *t*.

 A_{ft} is firm f's I/B/E/S-provided actual year t earnings.

 $CONNECTIONS_{aft} = \frac{\sum_{i} CONNECTIONS \#_{aft}^{i}}{INST_{OWNER} \#_{ft}}$ = the average number of connections between analyst *a* and the institutions holding stock, *f*, prior to analyst *a*'s forecast of *f*'s year *t* earnings.

CONNECTIONS $\#_{aft}^{i}$ = the number of stocks, excluding stock *f*, covered by analyst *a* and held by institution *i*, which invests in stock *f*, calculated using data prior to the time of F_{aft} .⁷

INST_OWNER $\#_{ft}$ = the number of institutional investors holding stock *f*.

⁶ If an analyst issues more than one forecast during the 90-day period, we retain only the earliest forecast. The absolute forecast error is scaled by actual earnings.

⁷ Institutional holdings used to construct *CONNECTIONS*_{*aft*} are from the calendar quarter preceding analyst *a*'s forecast of firm *f*'s year *t* earnings and analyst coverage of other companies is from the one-year period preceding the calendar quarter end used to measure institutional holdings.

Variables which control for factors that could affect accuracy of earnings forecasts generated by analysts include:

 $ACCURACY_{af,t-1}$ = lagged value of the forecast accuracy measure defined above.

 $FIRM\#_t^a$ = number of firms analyst *a* followed in the year ending with the date of F_{aft} .

INDUSTRY $\#_{at}$ = number of industries analyst *a* followed in the year ending with the date of F_{aft} .

 $FIRM_EXP_{aft}$ = number of years since the first year analyst *a* issued one-year ahead earnings forecasts for firm *f* up to the date of F_{aft} .

 $BSIZE_{at}$ = number of analysts employed by analyst *a*'s brokerage house or research firm in the year ending with the date of F_{aft} .

 $DAYS_{aft}$ = number of days between the date of F_{aft} and the most recent one-year ahead forecast of firm f's year t earnings preceding F_{aft} by analyst a.

 EPS_FREQ_{aft} = frequency of analyst *a*'s one-year ahead earnings forecasts for firm *f* in the one-year period prior to the date of F_{aft} .

 $HORIZON_{aft}$ = number of days between the date of F_{aft} and the end of fiscal year t.

All control variables are scaled to fall between 0 and 1 based on the equation below:

 $Scaled \ Control_{aft} = \frac{|Control_{aft}| - \min(|Control_{ft}|)}{\max(|Control_{ft}|) - \min(|Control_{ft}|)}$

To test H1, we also employ a piecewise regression in model (2) below, which allows us to calculate the sensitivity of *ACCURACY* to *CONNECTIONS* in each *CONNECTIONS* tercile.⁸

$$ACCURACY_{aft} = \beta_0 + \sum_{k=1}^{3} \beta_k^{Tercile} D_k^{Tercile} CONNECTION_{aft} + \sum_m \beta_m Control_m + \varepsilon_{aft}$$
(2)

where $D_k^{Tercile}$ is a (1/0) indicator variable equaling one for the kth *CONNECTIONS* tercile (1=lowest and 3=highest) and zero otherwise. Under H1, we expect $\beta_1^{Tercile} > \beta_3^{Tercile}$.

⁸ Tercile cut-off points are derived from the distribution of *CONNECTIONS*_{aft}.

3.2 Model for testing H2

If the sensitivity of *ACCURACY* to *CONNECTIONS* increases when buy-side analysts produce greater amounts of private information, we expect $\beta_1 > \beta_5$ in the regression model (3) below. On the other hand, if buy-side analyst selection of already-accurate sell-side analysts drives the relation between *ACCURACY* and *CONNECTIONS* then we expect the sensitivity of *ACCURACY* to *CONNECTIONS* to decrease when buy-side analysts produce more of their own private information, and we expect $\beta_1 < \beta_5$ in the regression model (3) below.

$$\begin{aligned} ACCURACY_{aft} &= \beta_0 + \beta_1 CONNECTIONS_{aft}^{High \ Opp} + \beta_2 CONNECTIONS_{aft}^{High \ Opp \ 2} \\ &+ \beta_3 CONNECTIONS_{aft}^{Med \ Opp} + \beta_4 CONNECTIONS_{aft}^{Med \ Opp \ 2} \\ &+ \beta_5 CONNECTIONS_{aft}^{Low \ Opp} + \beta_6 CONNECTIONS_{aft}^{Low \ Opp \ 2} \\ &+ \sum_m \beta_m Control_m + \varepsilon_{aft} \end{aligned}$$
(3)

where $CONNECTIONS_{aft}^{High \, Opp}$ is constructed by:

$$CONNECTIONS_{aft}^{High \ Opp} = \frac{\sum_{i} CONNECTIONS_{aft}^{i} \times D_{High \ Opp}^{i}}{INST_{OWNER_{ft}}}$$

where $D_{High OPP}^{i}$ denotes an institutional investor from which sell-side analysts have more opportunities to acquire useful information. $INST_OWNER\#_{ft}$ denotes the number of institutional investors holding stock *f* at time *t*. $CONNECTIONS_{aft}^{Med Opp}$ and $CONNECTIONS_{aft}^{Low Opp}$ are constructed in a similar fashion for medium and low opportunity institutional investors. Note that normalizing the high/medium/low opportunity connections variables by $INST_OWNER\#_{ft}$ ensures that they add up to $CONNECTIONS_{aft}^{Low}$.

As described when we introduced H3, our approach to identifying high, medium, and low opportunity institutional investors depends on an institution's reliance on public information (RPI). Following Kacperczyk & Seru (2007), we construct RPI for each institution within each

firm-year, as the R^2 from an institution-level regression of changes in the number of shares held in a given stock by a given institution on lagged changes in mean analyst recommendations. All institutional investors are ranked each quarter by their RPI into terciles. Institutions in the lowest (highest) RPI tercile are considered as relying the least (most) on public information and classified as high- (low-) opportunity institutions.

3.3 Model for testing H3 and H4

To test whether the sensitivity of forecast accuracy to connections increases with sell-side analyst demand for information from the buy side or with buy-side analyst demand for information from the sell-side, we employ the following regression model.

$$\begin{aligned} ACCURACY_{aft} &= \beta_0 + \beta_1 CONNECTIONS_{aft} + \beta_2 CONNECTIONS_{aft}^2 \\ &+ \beta_3 DEMAND + \beta_4 DEMAND \times CONNECTIONS_{aft} \\ &+ \beta_5 DEMAND \times CONNECTIONS_{aft}^2 + \sum_m \beta_m Control_m + \varepsilon_{aft} \end{aligned}$$
(4)

where *DEMAND* is a variable reflecting demand by sell-side (buy-side) analysts for information from the buy-side (sell-side) analysts with whom they are connected. The indicator variable, $POST_GS_t$, proxies for demand from the sell-side. $POST_GS_t$ equals 1 for forecasts issued after 2003, and 0 for forecasts issued before 2001.⁹ The variables, $FIRM_EXP_{aft}$ and $ACCURACY_{af,t-1}$, proxy for demand from the buy-side. $FIRM_EXP_{aft}$ equals the number of years in which analyst *a* issued one-year ahead earnings forecasts for stock *f* up to the current year. $ACCURACY_{af,t-1}$ equals the lagged value of the forecast accuracy measure defined above.

⁹ We exclude the regulatory period years, 2001-2003, from this analysis. Regulation Fair Disclosure was promulgated in October of 2000, the Sarbanes Oxley Act took effect in July of 2002, and the Global Analyst Research Settlement occurred in April of 2003.

If sell-side analyst demand drives the relation between *ACCURACY* and *CONNECTIONS* in model (4) above, then we expect $\beta_4 > 0$, when the demand proxy is *POST_GS_t*. Similarly, if buy-side demand drives the relation between *ACCURACY* and *CONNECTIONS*, then we expect $\beta_4 > 0$ when the demand proxy is either *FIRM_EXP_{aft}* or *ACCURACY_{af,t-1}*.

4 Sample selection

Our sample contains 137,835 analyst-firm-year observations from 1995 to 2012, including 3,980 unique firms and 7,615 unique analysts. We employ the following sample construction steps. We require each observation to have (from I/B/E/S): a one-year ahead EPS forecast issued between 1985 and 2012 and during the first 90 days following the prior year's earnings announcement, and an identifier that allows linkage to the CRSP database. If more than one analyst-firm-year observation exists in the same 90-day period, we keep only the earliest one. We also eliminate all firm-year observations with zero or missing institutional holdings data in the Thomson Reuters 13F database. We impose additional filters associated with data requirements needed to measure analyst characteristic control variables, such as the lagged forecast error. Finally, we require that each firm-year observation has coverage by more than one analyst. Our sample period begins in 1995, the first year when we can construct the RPI measure of Kacperczyk & Seru (2007) and ends in 2012 due to a warning from WRDS documenting severe coverage problems with Thomson Reuters 13F database after 2012.¹⁰ Nonetheless, in unreported supplemental tests, we extended the sample period to 2015 and our results and inferences remain robust.

¹⁰ The RPI measure is constructed for each institutions by regressing changes in the number of shares held in a given stock on lagged changes in mean analyst recommendations. The mean analyst recommendations data in IBES starts in the last quarter of 1993, but because four lags of changes in mean recommendations are needed, 1995 is the first year for which RPI can be constructed for each institutional investor.

5 Hypotheses Test Results

5.1 Descriptive statistics

Table 1 Panel A presents descriptive statistics for variables in our models, along with some variable components. Panel A shows that the distribution of absolute analyst-firm-year forecast error has a mean (median) of 0.795 (0.146). The *ACCURACY* variable used in our hypotheses tests scales $|FE_{aft}|$ to fall in a range from 0 to 1. The mean (median) of *ACCURACY* is 0.528 (0.550). Scaling all independent variables in this manner maintains the relative values of the variable, while allowing comparison across regression coefficients (Clement & Tse, 2005).

The *CONNECTIONS* variable indicates that, on average, analysts have 6.7 connections per institution holding a given stock followed by the analyst. This means that, on average, an institution holding a given stock followed by an analyst also holds, on average, 6.7 other stocks followed by that analyst, while the average analyst follows 16.7 firms. We attribute this seemingly high overlap between analyst coverage and institutional holding to the relatively broad based and diversified portfolios of stocks held by the average institutional investor. In fact, the mean (median) number of institutional investors holding a particular stock is 293 (213).

Panel A also shows that, in an average analyst-firm-year, a given analyst follows stocks in 3.8 different industries, has about 5 years of experience forecasting the earnings of each stock that she covers, works for a brokerage house or research firm employing 67 analysts, issues forecasts 4.6 days after the most recent forecast by any analyst following the same firm, has issued 6 one-year ahead earnings forecasts in the year prior to the time of F_{aft} for the same firm, and has a forecast horizon until the end of the fiscal year averaging 310 days. Panel A also shows the averages of the number of connections that the average analyst has with institutions in the

three RPI categories proxying for the opportunity to glean useful information from institutional investors.

Table 1 Panel B shows the univariate correlations among the variables used to test our hypotheses. Consistent with prior literature, our measure of relative within firm-year *ACCURACY* is significantly positively correlated with the prior year's *ACCURACY*, the analyst's firm-specific experience, and brokerage house size; and *ACCURACY* is negatively correlated with the number of days since the most recent preceding analyst forecast, forecast frequency, number of industries followed, and the horizon between the forecast and the upcoming annual earnings announcement date. *ACCURACY* is not significantly correlated with *CONNECTIONS*, which is consistent with a non-linear relation requiring a quadratic term in our multiple regression analysis. Both *CONNECTIONS* and analyst forecast *ACCURACY* are significantly greater after the global settlement.

5.2 Test of H1

Table 2 displays the results of our test of H1, which predicts that the accuracy of analyst a's forecast of firm f's earnings improves, to a point of diminishing returns, with the degree of connectedness between the analyst and institutional investors who hold stock f in their portfolios. For ease of presentation, the dependent variable (and, thus, each coefficient) is multiplied by 100. The significantly positive coefficient on the *CONNECTIONS* variable and the significantly negative coefficient on the square of the *CONNECTIONS* variable confirm H1's prediction. The p-values being less than 0.01 and the magnitudes of the coefficients suggest that they are statistically, as well as economically, significant. The economic significance is, perhaps, more apparent in column (4) where we perform a piecewise estimation of *CONNECTIONS* terciles.

The coefficient on *CONNECTIONS* in the lowest tercile of the variable is 9.464, which means that a one-standard deviation change in *CONNECTIONS* (untabulated, at 0.345) is associated with a 3.265% (0.345x9.464) change in the *ACCURACY* dependent variable. That represents 6.2% of mean *ACCURACY* (0.0327/0.528). The less significant coefficient on *CONNECTIONS* in the second tercile and the insignificant coefficient in the highest tercile indicate that, once the analyst's average amount of connections per institution reaches an optimum level, diminishing returns to additional connections become apparent. The coefficient on *CONNECTIONS* in the lowest terciles is significantly larger than that in the highest terciles, with a p-value (untabulated) of less than 0.001.

With reference to the control variables in model (1), the results in Table 2 indicate that *ACCURACY* significantly improves with the number of firms the analyst covers during the year prior to the forecast date and with the prior year's *ACCURACY* (i.e., the *ACCURACY* variable is relatively stable from one year to the next). The coefficient on the number of firms covered is opposite to results in prior research but should be interpreted with caution because of the mechanically very high univariate correlation between *CONNECTIONS* and coverage.¹¹ *ACCURACY* significantly declines with the number of industries the analyst follows, the number of days between the forecast date and the date of the most recent prior forecast, and the number of days between the forecast date and the end of the firm's fiscal year. These results are consistent with prior literature.

5.3 Test of H2

¹¹ When excluding *CONNECTIONS* and its squared term, the coefficient on the number of firms covered becomes insignificant.

Panel A of Table 3 displays the results of our tests of H2, which predicts that the benefits to sell-side analysts of connections with buy-side analysts intensify when the buy-side analysts' institutional investment firms have more private information; i.e., when greater opportunities for sell-side analysts to learn from buy-side analysts exist. As described above, our proxy for opportunity is the institutional investor's RPI, measured in the manner described in Kacperczyk & Seru, (2007). We divide the observations into three terciles (high, medium, and low opportunity) and estimate the coefficients of *CONNECTIONS* and its quadratic terms within each tercile. We predict that connections with high opportunity institutions reap greater rewards in terms of improved forecast accuracy than connections with medium and low opportunity institutions.

The results in Panel A of Table 3 are consistent with our prediction; i.e., the strength of the relation between *ACCURACY* and *CONNECTIONS* increases when analysts are connected with high opportunity institutions. The difference between the coefficient on connections with highest opportunity institutions and that on connections with lowest opportunity institutions is significant at a p-value of 0.052 (untabulated). The results also support our primary hypothesis that connections with buy-side analysts inform sell-side analyst research and improve sell-side analyst forecast accuracy. Furthermore, these results relieve the endogeneity concern that institutions choose to connect more with already-accurate analysts. If this was happening, *low opportunity* institutions and instead rely on public sources of information such as the research output produced by analysts. Such a preference suggests a stronger relation between connections with *low opportunity* institutions and *ACCURACY*, which is opposite to what we find.

In Panel B of Table 3, we provide an assessment of economic significance for our main effect by estimating Models (1) and (2) for the subsample of observations whose connections with low *RPI* institutions are in the bottom tercile of each firm-year group. Because Panel A of Table 3 indicates that the results concentrate in low RPI institutions and the relation between connections and accuracy is nonlinear, we expect larger than average economic significance for this subsample. For this subsample, the untabulated mean, median and standard deviation of *CONNECTIONS* with low *RPI* institutions (*CONNECTIONS^{High Opp}*) are 0.517, 0.559, and 0.423, respectively. From Column 2, the coefficient on *CONNECTIONS^{High Opp}* in the lowest tercile of the variable is 38.567, which means that a one-standard deviation change in *CONNECTIONS^{HighOpp}* is associated with a 16.31% (0.423×38.567) change in the *ACCURACY* dependent variable, representing 30.9% of mean *ACCURACY* (0.1631/0.528).

5.4 Test of H3

Table 4 displays the results of our test of H3, which predicts that variables proxying for the demand for information by sell-side analysts increases the impact of *CONNECTIONS* on *ACCURACY*.¹² The sell-side demand proxy is a dummy variable indicating whether the analyst-firm-year observation occurs after 2003; i.e., after Reg. FD and GARS constrained sell-side analysts' private communications with investment bankers in their own investment firms and with management of the firms the analysts follow. H3 predicts that after these changes in the information environment, sell-side analysts' connections with buy-side analysts are more accuracy-enhancing. Observations during the years of the changing regulatory environment (2001-2003) are omitted from this analysis.

¹² In tables 4 and 5, in the interest of brevity we report results only based on the quadratic specification.

Table 4 shows that the interaction *POST_GS* and *CONNECTIONS* is positive and significant. This supports the prediction from H3 that the regulatory shocks increased the impact of connections with buy-side analysts on sell-side analysts' forecast accuracy. In fact, the coefficients suggest that the *POST_GS* observations drive the results discussed in relation to Table 2 above. Overall, this result suggests that sell-side demand for information from buy-side analysts intensifies the importance of connections that inform sell-side analyst research and improve forecast accuracy.

5.5 Test of H4

Table 5 displays results from tests of H4, which predicts that variables proxying for the demand for information by buy-side analysts strengthens the relation between *CONNECTIONS* and *ACCURACY*. We proxy for this demand with (1) sell-side analyst experience in forecasting earnings of the subject firm and (2) lagged sell-side analyst forecast accuracy for the same firm.

In Table 5 we find no evidence that our proxies for buy-side interest in connecting with already-accurate sell-side analysts strengthen the relation between *ACCURACY* and *CONNECTIONS*. Neither the firm-specific experience proxy in column (1) nor the past-accuracy proxy in column (2) has a statistically significant interactive effect with the connections variable.

Overall, we believe that our tests of H2, H3, and H4 provide strong evidence that sellside interest in connecting with buy-side analysts in order to glean information that improves the quality of sell-side research reports drives the relation we find between *ACCURACY* and *CONNECTIONS*. The next section describes the results of additional robustness tests.

6 Additional Analyses

6.1 Another Test of Reverse Causality

As described in Section 1, a concern is that rather than sell-side analyst forecast accuracy improving due to connections with information-laden buy-side analysts, less information-laden buy-side analysts may choose to work with sell-side analysts with the best earnings forecast accuracy track records. In other words, the best sell-side analysts may learn nothing from connections with buy-side analysts. Instead, buy-side analysts may connect with the sell-side analysts from whom they can acquire the most information, and the flow of information may be one-directional; i.e., from the sell-side to the buy-side, with accuracy driving connections.

Results described in Sections 5.3 through 5.5 mitigate this concern. To further address this concern, we constrain the sample to sell-side analysts with less than four years of firm-specific experience. We argue that these analysts do not have enough of an accuracy track record to attract the interest of buy-side analysts in the companies they cover, which would then drive connections. In this sample, we expect that our firm-specific experience variable is not significant, while all of the other results still hold. Table 6 displays the results, which mirror the results testing H1 in Table 2, except that, as expected, the firm-specific experience variable is no longer significantly related to forecast accuracy. Thus, our inferences remain unchanged. This suggests that our main result is consistent with connections with buy-side analysts enhancing sell-side analyst forecast accuracy (not the other way around).

6.2 Alternative Measures of CONNECTIONS

Table 7 replicates our test of H1 using models (1) and (2) and two alternative proxies for the degree of connectedness between sell-side and buy-side analysts. The first alternative considers analyst a to have a connection with institution i that informs the analyst's forecasts of

firm f's year t earnings if i holds f and at least one other stock followed by a. We then divide the total number of a's connections by the number of institutions that hold f. Unlike our original *CONNECTIONS* variable, this proxy treats all institutions with whom analyst a is connected equally. The estimation results using this alternative measure of *CONNECTIONS* are presented in columns (1) and (2) of Table 7. The results are essentially the same as those presented in Table 2.

The second alternative is the same as our original method of estimating *CONNECTIONS* but with additional stringency. Recall that our original approach measures *CONNECTIONS* as the number of stocks other than *f* that *a* holds in common with each institution holding *f*, averaged across all institutions holding *f*. The additional stringency is that a stock other than *f* is only counted as a stock held in common with an institution, if all of the stocks covered by *a* account for at least 5% of the market value of the institution's market value. Essentially, this measure views an analyst as being connected with an institution only when the stocks commonly covered by the analyst and held by the institution account for a significant portion of the institutions' portfolio. The estimation results using this alternative measure of *CONNECTIONS* are presented in columns (3) and (4) of Table 7. Again, the results are essentially the same as those presented in Table 2. Overall, the results in Table 7 increase our confidence in the construct validity of our *CONNECTIONS* variable.

7 Conclusion

A plethora of research papers examine the impact of sell-side financial analyst research on the investment community (Ramnath et al. 2008b), while relatively few papers examine the role of buy-side analysts, working for institutional investors, the most important clients of the investment and boutique research firms that employ sell-side analysts (Brown et al. 2016). Ours is the first large-sample study to examine the impact of private buy-side analyst information on the quality of publicly available sell-side analyst research.

Sell-side analysts have substantial incentives to impress buy-side analysts working for various institutional investors, because buy-side votes for Institutional Investor's all-American team of sell-side analysts largely determine sell-side analyst compensation (Groysberg et al. 2011). Buy-side analysts have substantial incentives to enlist the services of sell-side analysts who add value to the information contained in buy-side analyst reports to their fund managers. The value added comes from industry expertise (Bradley, et al. 2017), connections with firm management (Green, Jame, Markov, & Subasi, 2014), and information gleaned from sell-side analysts' connections with other buy-side analysts (Brown et al. 2016).

Most prior academic research regarding the interactions between these two sophisticated groups of market participants focuses on the flow of information from sell-side analyst research into stock prices. Some studies focus on the flow of information from stock price changes into sell-side analyst research (e.g., Clement, Hales, & Xue 2011). Our study focuses directly on what sell-side analysts learn from connections with institutional investors. Our evidence of a non-linear relation between connections with institutional investors and sell-side analyst earnings forecast accuracy is consistent with these connections enhancing the quality of sell-side analyst research output and, hence, the quality of information impounded in capital asset prices, although up to a point of diminishing returns.

We recognize that buy-side analysts may invest effort in choosing the sell-side analysts whom they wish to engage, and this choice may depend on the accuracy of sell-side analyst earnings forecasts. At the same time, we hypothesize that the accuracy of sell-side analyst

earnings forecasts depends on the intensity of their connections with buy-side analysts. Our tests effectively untangle this endogeneity and focus on the flow of information from the buy-side to the sell-side. To the best of our knowledge, ours is the first study to show that sell-side analysts learn about the stocks they follow from connections with their buy-side counterparts.

References

Bernhardt, D., Wan, C., & Xiao, Z. (2016). The reluctant analyst. *Journal of Accounting Research*, *54*(4), 987-1040.

Bradley, D., Gokkaya, S., & Liu, X. (2017). Before an analyst becomes an analyst: Does industry experience matter? *The Journal of Finance*, 72(2), 751-792.

Bradshaw, M. T. (2011). Analysts' Forecasts: What do we know after decades of work? Working paper, Boston College, 2011. Available at http://papers.ssrn.com/sol3/papers.cfm?abstract id=1880339.

Brown, L. D. (2001). How important is past analyst forecast accuracy? *Financial Analysts Journal*, *57*(6), 44-49.

Brown, L. D., Call, A. C., Clement, M. B., & Sharp, N. Y. (2016). The activities of buy-side analysts and the determinants of their stock recommendations. *Journal of Accounting and Economics*, *62*(1), 139-156.

Bushee, B. J., Jung, M. J., & Miller, G. S. (2016). Do investors benefit from selective access to management? *Journal of Financial Research*.

Clement, M. B. (1999). Analyst forecast accuracy: Do ability, resources, and portfolio complexity matter? *Journal of Accounting and Economics*, 27(3), 285-303.

Clement, M. B., Hales, J., & Xue, Y. (2011). Understanding analysts' use of stock returns and other analysts' revisions when forecasting earnings. *Journal of Accounting and Economics*, *51*(3), 279-299.

Clement, Michael B., & Senyo Y. Tse. "Financial analyst characteristics and herding behavior in forecasting." *The Journal of finance* 60.1 (2005): 307-341.

Ertimur, Y., Sunder, J., & Sunder, S. V. (2007). Measure for measure: The relation between forecast accuracy and recommendation profitability of analysts. *Journal of Accounting Research*, *45*(3), 567-606.

Franck, A., & Kerl, A. (2013). Analyst forecasts and European mutual fund trading. *Journal of Banking & Finance*, *37*(8), 2677-2692.

Gleason, C. A., Bruce Johnson, W., & Li, H. (2013). Valuation model use and the price target performance of sell- side equity analysts. *Contemporary Accounting Research*, *30*(1), 80-115.

Green, T. C., Jame, R., Markov, S., & Subasi, M. (2014). Access to management and the informativeness of analyst research. *Journal of Financial Economics*, *114*(2), 239-255.

Greenwich Associates, 2010. U.S. equity research: Bulge bracket down, but not out. Greenwich

Report (June).

Groysberg, B., Healy, P., & Chapman, C. (2008). Buy-side vs. sell-side analysts' earnings forecasts. *Financial Analysts Journal*, 64(4), 25-39.

Groysberg, B., Healy, P. M., & Maber, D. A. (2011). What drives sell-side analyst compensation at high-status investment banks? *Journal of Accounting Research*, *49*(4), 969-1000.

Gu, Z., Li, G., Li, Z., & Yang, Y. G. (2016). Friends in need are friends indeed: An analysis of social ties between financial analysts and mutual fund managers. Working Paper.

Horton, J., Serafeim, G., & Wu, S. (2017). Career concerns of banking analysts. *Journal of Accounting and Economics*, 63(2), 231-252.

Irvine, P., Lipson, M., & Puckett, A. (2007). Tipping. *The Review of Financial Studies*, 20(3), 741-768.

Jacob, J., Lys, T. Z., & Neale, M. A. (1999). Expertise in forecasting performance of security analysts. *Journal of Accounting and Economics*, 28(1), 51-82.

Jacob, J., Rock, S., & Weber, D. P. (2008). Do non-investment bank analysts make better earnings forecasts?. *Journal of Accounting, Auditing & Finance*, 23(1), 23-61.

Kacperczyk, M., & Seru, A. (2007). Fund manager use of public information: New evidence on managerial skills. *The Journal of Finance*, 62(2), 485-528.

Maber, D., Groysberg, B., & Healy, P. (2015). Client service, compensation, and the sell-side analyst objective function: An empirical analysis of relational incentives in the investment-research industry. Cal Poly Orfalea School of Business Working Paper.

Merkley, K., Michaely, R., & Pacelli, J. (2017). Does the scope of the sell- side analyst industry matter? An examination of bias, accuracy, and information content of analyst reports. *The Journal of Finance*, *72*(3), 1285-1334.

Mikhail, M. B., Walther, B. R., & Willis, R. H. (2007). When security analysts talk, who listens? *The Accounting Review*, 82(5), 1227-1253.

Myring, M., & Wrege, W. (2011). Analysts earnings forecast accuracy and activity: A timeseries analysis. *Journal of Business & Economics Research (JBER)*, 7(5).

Pelletier, M. (2015). Five investing tips from a former sell-side analyst. Financial Post: http://business.financialpost.com/investing/investing-pro/five-investing-tips-from-a-former-sell-side-analyst/wcm/f270992a-284a-42dd-aec8-c1102175dd2d.

Ramnath, S., Rock, S., & Shane, P. B. (2008a). Financial analysts' forecasts and stock recommendations: A review of the research. *Foundations and Trends*® *in Finance*, *2*(4), 311-421.

Ramnath, S., Rock, S., & Shane, P. (2008b). The financial analyst forecasting literature: A taxonomy with suggestions for further research. *International Journal of Forecasting*, 24(1), 34-75.

Solomon, D., & Soltes, E. (2015). What are we meeting for? The consequences of private meetings with investors. *The Journal of Law and Economics*, *58*(2), 325-355.

Thomson Reuters 2009 IR Best Practices. https://issuu.com/irwebreport/docs/2009irpracticesstudy_executivesummary.

Table 1—Descriptive Statistics and Correlations

Variables	mean	p25	p50	p75	Standard Deviation
FE	0.795	0.053	0.146	0.408	5.571
FE	0.476	-0.107	0.000	0.231	5.608
ACCURACY	0.528	0.208	0.550	0.864	0.357
CONNECTIONS	6.680	3.908	5.852	8.267	4.615
INST_OWNER#	292.827	130.000	213.000	363.000	251.839
FIRM#	16.665	11.000	15.000	20.000	9.525
INDUSTRY#	3.825	2.000	3.000	5.000	2.663
FIRM_EXP	4.953	2.000	4.000	7.000	4.137
BSIZE	67.109	22.000	52.000	101.000	58.847
DAYS	4.589	0.000	0.000	3.000	12.525
EPS_FREQ	6.125	4.000	6.000	7.000	2.701
HORIZON	310.102	295.000	320.000	334.000	31.639
CONNECTIONS with high RPI institutions	0.092	0.019	0.049	0.106	0.150
CONNECTIONS with medium RPI institutions	0.453	0.174	0.325	0.565	0.485
CONNECTIONS with low RPI institutions	5.882	3.333	5.144	7.398	4.101

Panel A. Descriptive Statistics (137,835 analyst-firm-year observations)

Panel B. Correlations

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
CONNECTIONS (1)	1.000											
ACCURACY (2)	0.001	1.000										
	0.811											
Lag year ACCURACY (3)	-0.010	0.049	1.000									
	0.000	0.000										
FIRM# (4)	0.822	0.004	-0.016	1.000								
	0.000	0.192	0.000									
INDUSTRY# (5)	0.410	-0.012	-0.015	0.474	1.000							
	0.000	0.000	0.000	0.000								
FIRM_EXP (6)	0.136	0.007	-0.004	0.135	0.072	1.000						
	0.000	0.008	0.135	0.000	0.000							
BSIZE (7)	0.123	0.006	0.007	0.055	-0.060	0.033	1.000					
	0.000	0.042	0.006	0.000	0.000	0.000						
DAYS (8)	0.077	-0.012	-0.007	0.056	0.055	0.094	0.032	1.000				
	0.000	0.000	0.011	0.000	0.000	0.000	0.000					
EPS_FREQ (9)	-0.032	-0.022	-0.070	-0.039	-0.055	-0.014	0.080	-0.014	1.000			
	0.000	0.000	0.000	0.000	0.000	0.000	0.000	0.000				
HORIZON (10)	-0.062	-0.123	-0.007	-0.063	-0.056	-0.039	-0.024	-0.235	0.140	1.000		
	0.000	0.000	0.016	0.000	0.000	0.000	0.000	0.000	0.000			
CONNECTIONS with low RPI institutions (11)	0 004	0.002	-0.010	0 835	0 413	0 136	0 124	0.075	-0.033	-0.065	1.000	
institutions	0.000	0.002	-0.010	0.000	0.413	0.130	0.124	0.075	-0.033	-0.005	1.000	
POST GS (12)	0.000	0.000	0.000	0.000	0.000	-0.060	-0.010	- 0 080	-0 010	0.064	0.061	1.00
(12)	0.000	0.043	0.021	0.007	0.000	0.000	-0.010	0.000	0.000	0.004	0.001	1.00

Panel A reports summary statistics for our sample of 137,835 analyst-firm-year observations from 1995 to 2012, including 3,980 unique firms and 7,615 analysts. Panel B reports correlation coefficients (and associated p-values) among the main variables used in the analysis, where the variables are scaled among analysts for the same firm-year. For ease of interpretation, with the exception of *ACCURACY*, no variable in Panel A is scaled among analysts for the same firm-year.

We choose the absolute forecast errors for the earliest forecasts of all analysts forecasting firm f's year t earnings, where the forecasts occur during the 90 days post the announcement of firm f's year t-1 annual earnings. Based on these forecast errors, we compute ACCURACY as the difference between the maximum and analyst a's absolute forecast error, scaled by the range between the maximum and minimum. ACCURACY falls on a scale between zero (least accurate) and one (most accurate).

- $|FE_{aft}|$ = absolute error in analyst *a*'s earliest forecast, F_{aft} , for firm *f*'s year *t* earnings issued during the 90 days post the announcement of firm *f*'s year *t*-1 annual earnings, scaled by the absolute value of actual earnings.
- $CONNECTIONS_{aft}$ = analyst *a*'s average number of connections with institutional investors holding *f* as of the date of F_{aft} , defined as the number of stocks, other than *f*, covered by analyst *a* and held by institutions that invest in firm *f*, divided by the number of all institutions holding firm *f*. Institutional holdings used to construct this measure are from the calendar quarter preceding the date of F_{aft} , and analyst coverage of other companies is from the one year period that precedes the calendar quarter end used for institutional holding measurement.
- $FIRM \#_t^a$ = number of firms analyst *a* followed in the year ending with the date of F_{aft} .

INDUSTRY#_{at} = number of industries analyst a followed in the year ending with the date of F_{aft} .

 $FIRM_EXP_{aft}$ = number of years since the first year analyst *a* issued one-year ahead earnings forecasts for firm *f* up to the date of F_{aft} .

 $BSIZE_{at}$ = number of analysts employed by analyst a's brokerage house or research firm in the year ending with the date of F_{aff} .

 $DAYS_{aft}$ = number of days between the date of F_{aft} and the most recent one-year ahead forecast of firm f's year t earnings preceding F_{aft} by any analyst. EPS_FREQ_{aft} = frequency of analyst a's one-year ahead earnings forecasts for firm f in the one-year period prior to the date of F_{aft} .

 $HORIZON_{aft}$ = number of days between the date of F_{aft} and the end of fiscal year t.

 $CONNECTIONS_{aft}^{High \, Opp}$, $CONNECTIONS_{aft}^{Med \, Opp}$, and $CONNECTIONS_{aft}^{Low \, Opp}$ are measured the same way as the original CONNECTIONS variable except that they are constructed based on connections with subsets of institutions, i.e., high-, medium-, and low-opportunity institutions. To classify institutions into high-, medium-, and low-opportunity institutions, we measure an institution's reliance on public information (RPI). We construct RPI of an institution as the R² from an institution-level regression of changes in the number of shares held by that institution in a given stock on lagged changes in mean analyst recommendations. All institutional investors are ranked each quarter by their RPI into terciles. Institutions in the lowest RPI tercile are viewed as having the least reliance on public information and classified as high-opportunity institutions.

	(1)	(2)	(3)	(4)
Variables	Coeff (std. err.)	Coeff (std. err.)	Coeff (std. err.)	Coeff (std. err.)
CONNECTIONS	5.679***	5.166***		
	(1.176)	(1.266)		
CONNECTIONS ²	-5.723***	-5.911***		
	(1.217)	(1.188)		
Break down of CONNECTIONS	· · · ·	. ,		
Bottom CONNECTIONS Tercile			9.430***	9.464***
			(2.430)	(2.458)
Middle CONNECTIONS Tercile			2.558***	2.097**
			(0.738)	(0.867)
Top CONNECTIONS Tercile			0.484	-0.137
			(0.407)	(0.577)
Lagged ACCURACY		4.554***		4.565***
		(0.314)		(0.315)
FIRM#		0.942*		0.920
		(0.568)		(0.568)
INDUSTRY#		-1.648***		-1.699***
		(0.348)		(0.350)
FIRM_EXP		0.688**		0.704**
		(0.286)		(0.286)
BSIZE		0.251		0.269
		(0.345)		(0.347)
DAYS		-3.895***		-3.947***
		(0.298)		(0.297)
EPS_FREQ		-0.010		0.001
		(0.302)		(0.302)
HORIZON		-11.796***		-11.789***
		(0.274)		(0.275)
Constant	52.910***	58.341***	52.878***	58.282***
	(1.012)	(1.031)	(1.016)	(1.035)
Year Fixed Effects	YES	YES	YES	YES
Ν	137,835	137,835	137,835	137,835
Adjusted R-squared	0.21%	2.18%	0.20%	2.17%

Table 2—Earnings Forecast Accuracy and Connections

Table 2 examines the relation between sell side analysts' forecast accuracy and their connections with institutional investors. This table reports coefficient estimates from the following regressions: $ACCURACY_{aft} = \beta_0 + \beta_1 CONNECTIONS_{aft} + \beta_2 CONNECTIONS_{aft}^2 + \sum_m \beta_m Control_m + \varepsilon_{aft}$ (1) $ACCURACY_{aft} = \beta_0 + \sum_{k=1}^{3} \beta_k^{Tercile} D_k^{Tercile} CONNECTION_{aft} + \sum_m \beta_m Control_m + \varepsilon_{aft}$ (2) $D_k^{Tercile}$ is a (1/0) indicator variable equaling one for the kth CONNECTIONS tercile (1=lowest and 3=highest) and zero otherwise. All other variables are defined as in Table 1. In this table, all continuous variables are scaled to fall between 0 and 1 for the same firm-year. Standard errors are clustered by analyst and are presented in parentheses. ***, **, * denote statistical significance at the 1%, 5%, and 10% significance level, respectively.

Table 3— Earnings Forecast Accuracy and Connections Stratified by Opportunities for Analysts to Learn

Panel A. All observations

Variables	coeff (std. err.)
CONNECTIONS High Opp	5.719***
	(1.709)
CONNECTIONS High Opp2	-3.888***
	(1.462)
CONNECTIONS Med Opp	0.053
	(1.587)
CONNECTIONS Med Opp2	-1.640
	(1.459)
CONNECTIONS Low Opp	1.470
	(1.202)
CONNECTIONS Low Opp2	-2.667**
	(1.176)
Lagged ACCURACY	4.533***
	(0.313)
FIRM#	0.345
	(0.596)
INDUSTRY#	-1.550***
	(0.345)
FIRM_EXP	0.729**
Dalas	(0.286)
BSIZE	0.237
DAVC	(0.345)
DAYS	-3.812***
EDS EDEO	(0.298)
EFS_I'KEQ	-0.019
HORIZON	(0.302)
HORZON	(0.275)
Constant	(0.27 <i>3)</i> 58 473***
	(1.033)
Ν	137,835
Year Fixed Effects	YES
Adjusted R-squared	2.20%

	(1)	(2)
Variables	coeff (std. err.)	coeff (std. err.)
CONNECTIONS ^{High Opp}	9.576***	
	(1.950)	
CONNECTIONS ^{High Opp2}	-8.754***	
	(1.896)	
Break down of CONNECTIONS ^{High Opp}		
Bottom CONNECTIONS ^{High Opp} Tercile		38.567***
		(11.676)
Middle CONNECTIONS ^{High Opp} Tercile		3.371***
		(0.754)
Top CONNECTIONS ^{High Opp} Tercile		0.784
-		(0.525)
Lagged ACCURACY	3.598***	3.614***
	(0.440)	(0.440)
FIRM#	1.056**	1.043**
	(0.529)	(0.529)
INDUSTRY#	-1.074**	-1.077**
	(0.468)	(0.468)
FIRM_EXP	1.618***	1.622***
	(0.437)	(0.438)
BSIZE	1.280***	1.272***
	(0.452)	(0.452)
DAYS	-2.616***	-2.628***
	(0.457)	(0.457)
EPS_FREQ	-0.241	-0.239
	(0.452)	(0.453)
HORIZON	-7.033***	-7.028***
	(0.407)	(0.407)
Constant	52.104***	52.211***
	(1.850)	(1.846)
Ν	37,563	37,563
Year Fixed Effects	YES	YES
Adjusted R-squared	1.41%	1.41%

Panel B. Observations where analysts' connections with institutions are in the High Opportunity tercile

Table 3 examines whether the sensitivity of forecast accuracy to connections increases when sell-side analysts have greater opportunities to learn private information from their connections with institutional investors. Panel A reports results from the following regression model.

$$\begin{aligned} ACCURACY_{aft} &= \beta_0 + \beta_1 CONNECTIONS_{aft}^{High\,Opp} + \beta_2 CONNECTIONS_{aft}^{High\,Opp\,2} \\ &+ \beta_3 CONNECTIONS_{aft}^{Med\,Opp} + \beta_4 CONNECTIONS_{aft}^{Med\,Opp\,2} \\ &+ \beta_5 CONNECTIONS_{aft}^{Low\,Opp} + \beta_5 CONNECTIONS_{aft}^{Low\,Opp\,2} \\ &+ \sum_m \beta_m Control_m + \varepsilon_{aft} \end{aligned}$$
(3)

where $CONNECTIONS_{aft}^{High \, opp}$ is measured the same way as the original CONNECTIONS variable except that it is constructed only based on connections with high-opportunity institutions. $CONNECTIONS_{aft}^{Med \, Opp}$ and $CONNECTIONS_{aft}^{Low \, Opp}$ are constructed in a similar fashion based on connections with medium and lowopportunity institutions, respectively. To classify institutions into high-, medium-, and low-opportunity institutions, we measure an institution's reliance on public information (RPI). We construct RPI of an institution as the R² from an institution-level regression of changes in the number of shares held in a given stock by a given institution on lagged changes in mean analyst recommendations. All institutional investors are ranked each quarter by their RPI into terciles. Institutions in the lowest RPI tercile are classified as highopportunity institutions. All other variables are defined as in Table 1. In this table, all continuous variables are scaled to fall between 0 and 1 for the same firm-year. Panel B reports results from estimation of Models 1 and 2 from Table 2 on a subset of our sample including only the observations where analyst connections with High Opportunity institutions, i.e., those in the low RPI group, are in the bottom tercile of the same firm-year group. Standard errors are clustered by analyst and are presented in parentheses. ***, **, * denote statistical significance at the 1%, 5%, and 10% significance level, respectively.

Variables	coeff (std. err.)
CONNECTIONS	-0.462
	(2.504)
CONNECTIONS ²	-2.358
	(2.463)
CONNECTIONS× POST_GS	7.378***
	(2.845)
CONNECTIONS ² × POST_GS	-5.106*
	(2.866)
POST GS	1.079
-	(1.119)
Lagged ACCURACY	5.029***
	(0.337)
FIRM#	1.159*
	(0.616)
INDUSTRY#	-1.474***
	(0.371)
FIRM_EXP	0.809***
	(0.306)
BSIZE	-0.063
	(0.366)
DAYS	-3.959***
	(0.317)
Constant	59.092***
	(1.092)
Ν	120,983
Year Fixed Effects	YES
Adjusted R-squared	0.0209

Table 4— The Relation between Earnings Forecast Accuracy and Connections Pre- andPost-Exogenous Regulatory Changes

Table 4 examines whether the sensitivity of forecast accuracy to connections increases with sell-side analyst demand for information following the regulatory period that includes Reg. FD and GARS. This table reports coefficient estimates from the following regression.

$$\begin{aligned} ACCURACY_{aft} &= \beta_0 + \beta_1 CONNECTIONS_{aft} + \beta_2 CONNECTIONS_{aft}^2 \\ &+ \beta_3 POST_GS + \beta_4 POST_GS \times CONNECTIONS_{aft} \\ &+ \beta_5 POST_GS \times CONNECTIONS_{aft}^2 + \sum_m \beta_m Control_m + \varepsilon_{aft} \end{aligned}$$
(4)

 $POST_GS_t = 1$ for analyst forecasts issued after 2003, and 0 for forecasts issued before 2001. $FIRM_EXP_{aft} =$ number of years since the first year analyst *a* issued one-year ahead earnings forecasts for firm *f* up to current year.

*LAGGED_ACCURACY*_{aft}=one year lagged value of the *ACCURACY* variable.

All other variables are defined as in Table 1. Observations during the years of the changing regulatory environment (2001-2003) are omitted. In this table, all continuous variables are scaled to fall between 0 and 1 for the same firm-year. Standard errors are clustered by analyst and are presented in parentheses. ***, **, ** denote statistical significance at the 1%, 5%, and 10% significance level, respectively.

		(1)	(2)
		coeff (std. err.)	coeff (std. err.)
Variables	Demand =	FIRM_EXP	Lagged ACCURACY
CONNECTIONS		5.919***	5.950***
		(1.679)	(2.183)
CONNECTIONS ²		-6.287***	-6.122***
		(1.642)	(2.231)
CONNECTIONS × DEMAND		-2.137	-1.476
		(2.826)	(3.183)
CONNECTIONS ² × DEMAND		1.234	0.374
		(2.723)	(3.243)
Lagged ACCURACY		4.553***	5.076***
		(0.314)	(0.593)
FIRM#		0.941*	0.942*
		(0.568)	(0.568)
INDUSTRY#		-1.651***	-1.650***
		(0.347)	(0.348)
FIRM_EXP		1.234**	0.688**
		(0.566)	(0.286)
BSIZE		0.255	0.251
		(0.345)	(0.345)
DAYS		-3.895***	-3.894***
		(0.298)	(0.298)
Constant		58.147***	58.070***
		(1.045)	(1.065)
Ν		137,835	137,835
Year Fixed Effects		YES	YES
Adjusted R-squared		0.0218	0.0218

Table 5— The Relation between Earnings Forecast Accuracy and Connections by Buy-side Analyst Demand

Table 5 examines whether the sensitivity of forecast accuracy to connections increases with buy-side analyst demand for information. This table reports coefficient estimates from the following regressions.

 $\begin{aligned} ACCURACY_{aft} &= \beta_0 + \beta_1 CONNECTIONS_{aft} + \beta_2 CONNECTIONS_{aft}^2 + \beta_3 DEMAND + \beta_4 DEMAND \times CONNECTIONS_{aft}^2 + \beta_5 DEMAND \times CONNECTIONS_{aft}^2 + \sum_m \beta_m Control_m + \varepsilon_{aft} \end{aligned}$ (4)

We employ the following variables as proxies for buy-side analyst DEMAND:

 $FIRM_EXP_{aft}$ = number of years since the first year analyst *a* issued one-year ahead earnings forecasts for firm *f* up to current year.

*LAGGED_ACCURACY*_{aft}=one year lagged value of the *ACCURACY* variable.

All other variables are defined as in Table 1. In this table, all continuous variables are scaled to fall between 0 and 1 for the same firm-year. Standard errors are clustered by analyst and are presented in parentheses. ***, **, * denote statistical significance at the 1%, 5%, and 10% significance level, respectively.

	(1)	(2)
Variables	Coeff (std. err.)	Coeff (std. err.)
CONNECTIONS	4.665***	
	(1.744)	
CONNECTIONS ²	-5.204***	
	(1.666)	
Break down of CONNECTIONS		
Bottom CONNECTIONS Tercile		7.336**
		(3.428)
Middle CONNECTIONS Tercile		1.853
		(1.201)
Top CONNECTIONS Tercile		-0.046
		(0.799)
Lagged ACCURACY	4.660***	4.672***
	(0.436)	(0.437)
FIRM#	1.298	1.275
	(0.796)	(0.797)
INDUSTRY#	-1.888***	-1.926***
	(0.472)	(0.473)
FIRM_EXP	-0.158	-0.147
	(0.516)	(0.516)
BSIZE	0.259	0.262
	(0.462)	(0.463)
DAYS	-4.357***	-4.399***
	(0.433)	(0.432)
EPS_FREQ	-0.131	-0.124
	(0.415)	(0.416)
HORIZON	-11.482***	-11.472***
	(0.382)	(0.383)
Constant	57.533***	57.515***
	(1.526)	(1.531)
Year Fixed Effects	YES	YES
Ν	67,650	67,650
Adjusted R-squared	2.04%	2.03%

Table 6—Another Test for Reverse Causality

This table replicates the analysis from specifications (2) and (4) of Table 2 on the subset of our sample observations that include only the first three years since an analyst covered the same firm in the IBES database. Standard errors are clustered by analyst and are presented in parentheses. ***, **, * denote statistical significance at the 1%, 5%, and 10% significance level, respectively.

	(1)	(2)	(3)	(4)
Variables	Meas	sure 1	Meas	sure 2
variables	Coeff (std. err.)	Coeff (std. err.)
CONNECTIONS	5.535***		3.757***	
	(1.182)		(1.127)	
CONNECTIONS ²	-4.370***		-4.929***	
	(1.145)		(1.176)	
Break down of CONNECTIONS				
Bottom CONNECTIONS Tercile		2.421**		131.865***
		(0.969)		(43.737)
Middle CONNECTIONS Tercile		2.127***		3.107***
		(0.428)		(1.010)
Top CONNECTIONS Tercile		1.007***		-0.752**
		(0.362)		(0.341)
Lagged ACCURACY	4.558***	4.558***	4.550***	4.555***
	(0.315)	(0.315)	(0.314)	(0.315)
FIRM#	-0.002	-0.009	0.820**	0.795**
	(0.419)	(0.420)	(0.402)	(0.403)
INDUSTRY#	-1.694***	-1.706***	-1.606***	-1.632***
	(0.348)	(0.348)	(0.346)	(0.348)
FIRM_EXP	0.668**	0.676**	0.762***	0.770***
	(0.286)	(0.286)	(0.286)	(0.286)
BSIZE	0.145	0.137	0.238	0.259
	(0.345)	(0.345)	(0.344)	(0.344)
DAYS	-3.944***	-3.966***	-3.906***	-3.923***
	(0.297)	(0.297)	(0.298)	(0.297)
EPS_FREQ	0.004	0.005	-0.010	-0.009
	(0.302)	(0.302)	(0.302)	(0.302)
HORIZON	-11.807***	-11.799***	-11.791***	-11.792***
	(0.275)	(0.275)	(0.274)	(0.275)
Constant	57.987***	58.220***	58.735***	58.672***
	(1.042)	(1.041)	(1.028)	(1.029)
Ν	137,835	137,835	137,835	137,835
Year Fixed Effects	YES	YES	YES	YES
Adjusted R-squared	0.0217	0.0217	0.0217	0.0217

Table 7—Alternative Measures for Connections

This table replicates the analysis from specifications (2) and (4) of Table 2 with two alternative measures of the *CONNECTIONS* variable. The first alternative version of the *CONNECTIONS* variable is measured as the number of institutions that invest in both firm *j* and at least one other company followed by analyst *i*, divided by the number of all institutions holding firm *j*. The second alternative version of the *CONNECTIONS* variable is measured the same way as the original *CONNECTIONS* variable except that it considers only connections with institutions where the value of all stocks covered by an analyst as a percentage of an institution's portfolio value is above 5 percent of the institution's portfolio value. In other words, *CONNECTIONS* is calculated only for analyst-institution pairs where the aforementioned percentage is above 5 percent. Standard errors are clustered by analyst and are presented in parentheses. ***, **, * denote statistical significance at the 1%, 5%, and 10% significance level, respectively.

CFR working paper series



CFR working papers are available for download from www.cfr-cologne.de.

наrdcopies can be ordered from: centre for rinancial Research (сгя), Albertus Magnus Platz, 50923 коеln, Germany.

2017

No.	Author(s)	Title
17-04	G. Cici, P.B. Shane, Y. S. Yang	Do Connections with Buy-Side Analysts Inform Sell-Side Analyst Research?
17-03	G. Cici, S. Gibson, R. Moussawi	Explaining and Benchmarking Corporate Bond Returns
17-02	S. Jaspersen, P. Limbach	Knowing Me, Knowing You? Similarity to the CEO and Fund Managers' Investment Decisions
17-01	J. Grammig, EM. Küchlin	A two-step indirect inference approach to estimate the long-run risk asset pricing model

No.	Author(s)	Title
16-12	A.Betzer, M. Ibel, H.S. Lee, P. Limbach, J.M. Salas	Are Generalists Beneficial to Corporate Shareholders? Evidence from Sudden Deaths
16-11	P. Limbach, M. Schmid, M. Scholz-Daneshgari	Do CEOs Matter? Corporate Performance and the CEO Life Cycle
16-10	V. Agarwal, R. Vashishtha, M. Venkatachalam	Mutual fund transparency and corporate myopia
16-09	MA. Göricke	Do Generalists Profit from the Fund Families' Specialists? Evidence from Mutual Fund Families Offering Sector Funds
16-08	S. Kanne, O. Korn, M.Uhrig-Homburg	Stock Illiquidity, Option Prices and Option Returns
16-07	S. Jaspersen	Market Power in the Portfolio: Product Market Competition and Mutual Fund Performance
16-06	O. Korn, MO. Rieger	Hedging With Regret
16-05	E. Theissen, C. Westheide	Call of Duty: Designated Market Maker Participation in Call Auctions
16-04	P. Gomber, S. Sagade, E. Theissen, M.C. Weber, C. Westheide	Spoilt for Choice: Order Routing Decisions in Fragmented Equity Markets
16-03	T.Martin, F. Sonnenburg	Managerial Ownership Changes and Mutual Fund Performance

16-02	A.Gargano, A. G. Rossi, R. Wermers	The Freedom of Information Act and the Race Towards Information Acquisition
16-01	G. Cici, S. Gibson, C. Rosenfeld	Cross-Company Effects of Common Ownership: Dealings Between Borrowers and Lenders With a Common Blockholder
2015		
No.	Author(s)	Title
15-17	O. Korn, L. Kuntz	Low-beta Strategies
15-16	D. Blake, A.G. Rossi, A. Timmermann, I. Tonks, R. Wermers	Network Centrality and Pension Fund Performance
15-15	S. Jank, E. Smajbegovic	Dissecting Short-Sale Performance: Evidence from Large Position Disclosures
15-14	M. Doumet, P. Limbach, E. Theissen	Ich bin dann mal weg: Werteffekte von Delistings deutscher Aktiengesellschaften nach dem Frosta-Urteil
15-13	G. Borisova, P.K. Yadav	Government Ownership, Informed Trading and Private Information
15-12	V. Agarwal, G.O. Aragon, Z. Shi	Funding Liquidity Risk of Funds of Hedge Funds: Evidence from their Holdings
15-11	L. Ederington, W. Guan, P.K. Yadav	Dealer spreads in the corporate Bond Market: Agent vs. Market-Making Roles
15-10	J.R. Black, D. Stock, P.K. Yadav	The Pricing of Different Dimensions of Liquidity: Evidence from Government Guaranteed Bank Bonds
15-09	V. Agarwal, H. Zhao	Interfund lending in mutual fund families: Role of internal capital markets
15-08	V. Agarwal, T. C. Green, H. Ren	Alpha or Beta in the Eye of the Beholder: What drives Hedge Fund Flows?
15-07	V. Agarwal, S. Ruenzi, F. Weigert	Tail risk in hedge funds: A unique view from portfolio holdings
15-06	C. Lan, F. Moneta, R. Wermers	Mutual Fund Investment Horizon and Performance
15-05	L.K. Dahm, C. Sorhage	Milk or Wine: Mutual Funds' (Dis)economies of Life
15-04	A. Kempf, D. Mayston, M. Gehde-Trapp, P. K. Yadav	Resiliency: A Dynamic View of Liquidity
15-03	V. Agarwal, Y. E. Arisoy, N. Y. Naik	Volatility of Aggregate Volatility and Hedge Funds Returns
15-02	G. Cici, S. Jaspersen, A. Kempf	Speed of Information Diffusion within Fund Families
15-01	M. Baltzer, S. Jank, E. Smajlbegovic	Who trades on momentum?
2014		
No.	Author(s)	Title
14-15	M. Baltzer, S. Jank, E.Smajlbegovic	Who Trades on Monumentum?

14-14	G. Cici, L. K. Dahm, A. Kempf	Trading Efficiency of Fund Families: Impact on Fund Performance and Investment Behavior
14-13	V. Agarwal, Y. Lu, S. Ray	Under one roof: A study of simultaneously managed hedge funds and funds of hedge funds
14-12	P. Limbach, F. Sonnenburg	Does CEO Fitness Matter?
14-11	G. Cici, M. Gehde-Trapp, M. Göricke, A. Kempf	What They Did in their Previous Life: The Investment Value of Mutual Fund Managers' Experience outside the Financial Sector
14-10	O. Korn, P. Krischak, E. Theissen	Illiquidity Transmission from Spot to Futures Markets
14-09	E. Theissen, L. S. Zehnder	Estimation of Trading Costs: Trade Indicator Models Revisited
14-08	C. Fink, E. Theissen	Dividend Taxation and DAX Futures Prices
14-07	F. Brinkmann, O. Korn	Risk-adjusted Option-implied Moments
14-06	J. Grammig, J. Sönksen	Consumption-Based Asset Pricing with Rare Disaster Risk
14-05	J. Grammig, E. Schaub	Give me strong moments and time – Combining GMM and SMM to estimate long-run risk asset pricing
14-04	C. Sorhage	Outsourcing of Mutual Funds' Non-core Competencies
14-03	D. Hess, P. Immenkötter	How Much Is Too Much? Debt Capacity And Financial Flexibility
14-02	C. Andres, M. Doumet, E. Fernau, E. Theissen	The Lintner model revisited: Dividends versus total payouts
14-01	N.F. Carline, S. C. Linn, P. K. Yadav	Corporate Governance and the Nature of Takeover Resistance
2013		
No.	Author(s)	Title
13-11	R. Baule, O. Korn, S. Saßning	Which Beta is Best? On the Information Content of Option-implied Betas
13-10	V. Agarwal, L. Ma, K. Mullally	Managerial Multitasking in the Mutual Fund Industry
13-09	M. J. Kamstra, L.A. Kramer, M.D. Levi, R. Wermers	Seasonal Asset Allocation: Evidence from Mutual Fund Flows
13-08	F. Brinkmann, A. Kempf, O. Korn	Forward-Looking Measures of Higher-Order Dependencies with an Application to Portfolio Selection
13-07	G. Cici, S. Gibson, Y. Gunduz, J.J. Merrick, Jr.	Market Transparency and the Marking Precision of Bond Mutual Fund Managers

13-06 S. Bethke, M. Gehde-Trapp, A. Kempf Investor Sentiment, Flight-to-Quality, and Corporate Bond Comovement
13-05 P. Schuster, M. Trapp, M. Uhrig-Homburg The Term Structure of Bond Liquidity

13-03	V. Agarwal, V. Nanda, S.Ray	Institutional Investment and Intermediation in the Hedge Fund
		Industry
13-02	C. Andres, A. Betzer, M. Doumet, E. Theissen	Open Market Share Repurchases in Germany: A Conditional Event Study Approach
13-01	J. Gaul, E. Theissen	A Partially Linear Approach to Modelling the Dynamics of Spot and Futures Price
2012		
No.	Author(s)	Title
12-12	M. Gehde-Trapp, Y. Gündüz, J. Nasev	The liquidity premium in CDS transaction prices: Do frictions matter?
12-11	Y. Wu, R. Wermers, J. Zechner	Governance and Shareholder Value in Delegated Portfolio Management: The Case of Closed-End Funds
12-10	M. Trapp, C. Wewel	Transatlantic Systemic Risk
12-09	G. Cici, A. Kempf, C. Sorhage	Do Financial Advisors Provide Tangible Benefits for Investors? Evidence from Tax-Motivated Mutual Fund Flows
12-08	S. Jank	Changes in the composition of publicly traded firms: Implications for the dividend-price ratio and return predictability
12-07	G. Cici, C. Rosenfeld	A Study of Analyst-Run Mutual Funds: The Abilities and Roles of Buy-Side Analysts
12-06	A. Kempf, A. Pütz, F. Sonnenburg	Fund Manager Duality: Impact on Performance and Investment Behavior
12-05	L. Schmidt, A. Timmermann, R. Wermers	Runs on Money Market Mutual Funds
12-04	R. Wermers	A matter of style: The causes and consequences of style drift
12-03	C. Andres, A. Betzer, I. van den Bongard, C. Haesner, E. Theissen	Dividend Announcements Reconsidered: Dividend Changes versus Dividend Surprises
12-02	C. Andres, E. Fernau, E. Theissen	Should I Stay or Should I Go? Former CEOs as Monitors
12-01	L. Andreu, A. Pütz	Choosing two business degrees versus choosing one: What does it tell about mutual fund managers' investment behavior?

No.	Author(s)	Title
11-16	V. Agarwal, JP. Gómez, R. Priestley	Management Compensation and Market Timing under Portfolio Constraints
11-15	T. Dimpfl, S. Jank	Can Internet Search Queries Help to Predict Stock Market Volatility?
11-14	P. Gomber, U. Schweickert, E. Theissen	Liquidity Dynamics in an Electronic Open Limit Order Book: An Event Study Approach

11-13	D. Hess, S. Orbe	Irrationality or Efficiency of Macroeconomic Survey Forecasts? Implications from the Anchoring Bias Test
11-12	D. Hess, P. Immenkötter	Optimal Leverage, its Benefits, and the Business Cycle
11-11	N. Heinrichs, D. Hess, C. Homburg, M. Lorenz, S. Sievers	Extended Dividend, Cash Flow and Residual Income Valuation Models – Accounting for Deviations from Ideal Conditions
11-10	A. Kempf, O. Korn, S. Saßning	Portfolio Optimization using Forward - Looking Information
11-09	V. Agarwal, S. Ray	Determinants and Implications of Fee Changes in the Hedge Fund Industry
11-08	G. Cici, LF. Palacios	On the Use of Options by Mutual Funds: Do They Know What They Are Doing?
11-07	V. Agarwal, G. D. Gay, L. Ling	Performance inconsistency in mutual funds: An investigation of window-dressing behavior
11-06	N. Hautsch, D. Hess, D. Veredas	The Impact of Macroeconomic News on Quote Adjustments, Noise, and Informational Volatility
11-05	G. Cici	The Prevalence of the Disposition Effect in Mutual Funds' Trades
11-04	S. Jank	Mutual Fund Flows, Expected Returns and the Real Economy
11-03	G.Fellner, E.Theissen	Short Sale Constraints, Divergence of Opinion and Asset Value: Evidence from the Laboratory
11-02	S.Jank	Are There Disadvantaged Clienteles in Mutual Funds?
11-01	V. Agarwal, C. Meneghetti	The Role of Hedge Funds as Primary Lenders

2	01	.0
---	----	----

No.	Author(s)	Title
10-20	G. Cici, S. Gibson, J.J. Merrick Jr.	Missing the Marks? Dispersion in Corporate Bond Valuations Across Mutual Funds
10-19	J. Hengelbrock, E. Theissen, C. Westheide	Market Response to Investor Sentiment
10-18	G. Cici, S. Gibson	The Performance of Corporate-Bond Mutual Funds: Evidence Based on Security-Level Holdings
10-17	D. Hess, D. Kreutzmann, O. Pucker	Projected Earnings Accuracy and the Profitability of Stock Recommendations
10-16	S. Jank, M. Wedow	Sturm und Drang in Money Market Funds: When Money Market Funds Cease to Be Narrow
10-15	G. Cici, A. Kempf, A. Puetz	The Valuation of Hedge Funds' Equity Positions
10-14	J. Grammig, S. Jank	Creative Destruction and Asset Prices
10-13	S. Jank, M. Wedow	Purchase and Redemption Decisions of Mutual Fund Investors and the Role of Fund Families
10-12	S. Artmann, P. Finter, A. Kempf, S. Koch, E. Theissen	The Cross-Section of German Stock Returns: New Data and New Evidence

10-11	M. Chesney, A. Kempf	The Value of Tradeability
10-10	S. Frey, P. Herbst	The Influence of Buy-side Analysts on Mutual Fund Trading
10-09	V. Agarwal, W. Jiang, Y. Tang, B. Yang	Uncovering Hedge Fund Skill from the Portfolio Holdings They Hide
10-08	V. Agarwal, V. Fos, W. Jiang	Inferring Reporting Biases in Hedge Fund Databases from Hedge Fund Equity Holdings
10-07	V. Agarwal, G. Bakshi, J. Huij	Do Higher-Moment Equity Risks Explain Hedge Fund Returns?
10-06	J. Grammig, F. J. Peter	Tell-Tale Tails: A data driven approach to estimate unique market information shares
10-05	K. Drachter, A. Kempf	Höhe, Struktur und Determinanten der Managervergütung- Eine Analyse der Fondsbranche in Deutschland
10-04	J. Fang, A. Kempf, M. Trapp	Fund Manager Allocation
10-03	P. Finter, A. Niessen- Ruenzi, S. Ruenzi	The Impact of Investor Sentiment on the German Stock Market
10-02	D. Hunter, E. Kandel, S. Kandel, R. Wermers	Mutual Fund Performance Evaluation with Active Peer Benchmarks
10-01	S. Artmann, P. Finter, A. Kempf	Determinants of Expected Stock Returns: Large Sample Evidence from the German Market

No.	Author(s)	Title
09-17	E. Theissen	Price Discovery in Spot and Futures Markets: A Reconsideration
09-16	М. Тгарр	Trading the Bond-CDS Basis – The Role of Credit Risk and Liquidity
09-15	A. Betzer, J. Gider, D.Metzger, E. Theissen	Strategic Trading and Trade Reporting by Corporate Insiders
09-14	A. Kempf, O. Korn, M. Uhrig-Homburg	The Term Structure of Illiquidity Premia
09-13	W. Bühler, M. Trapp	Time-Varying Credit Risk and Liquidity Premia in Bond and CDS Markets
09-12	W. Bühler, M. Trapp	Explaining the Bond-CDS Basis – The Role of Credit Risk and Liquidity
09-11	S. J. Taylor, P. K. Yadav, Y. Zhang	Cross-sectional analysis of risk-neutral skewness
09-10	A. Kempf, C. Merkle, A. Niessen-Ruenzi	Low Risk and High Return – Affective Attitudes and Stock Market Expectations
09-09	V. Fotak, V. Raman, P. K. Yadav	Naked Short Selling: The Emperor`s New Clothes?
09-08	F. Bardong, S.M. Bartram, P.K. Yadav	Informed Trading, Information Asymmetry and Pricing of Information Risk: Empirical Evidence from the NYSE
09-07	S. J. Taylor , P. K. Yadav, Y. Zhang	The information content of implied volatilities and model-free volatility expectations: Evidence from options written on

		individual stocks
09-06	S. Frey, P. Sandas	The Impact of Iceberg Orders in Limit Order Books
09-05	H. Beltran-Lopez, P. Giot, J. Grammig	Commonalities in the Order Book
09-04	J. Fang, S. Ruenzi	Rapid Trading bei deutschen Aktienfonds: Evidenz aus einer großen deutschen Fondsgesellschaft
09-03	A. Banegas, B. Gillen, A. Timmermann, R. Wermers	The Cross-Section of Conditional Mutual Fund Performance in European Stock Markets
09-02	J. Grammig, A. Schrimpf, M. Schuppli	Long-Horizon Consumption Risk and the Cross-Section of Returns: New Tests and International Evidence
09-01	O. Korn, P. Koziol	The Term Structure of Currency Hedge Ratios
2008		
No.	Author(s)	Title
08-12	U. Bonenkamp, C. Homburg, A. Kempf	Fundamental Information in Technical Trading Strategies
08-11	O. Korn	Risk Management with Default-risky Forwards
08-10	J. Grammig, F.J. Peter	International Price Discovery in the Presence of Market Microstructure Effects
08-09	C. M. Kuhnen, A. Niessen	Public Opinion and Executive Compensation
08-08	A. Pütz, S. Ruenzi	Overconfidence among Professional Investors: Evidence from Mutual Fund Managers
08-07	P. Osthoff	What matters to SRI investors?
08-06	A. Betzer, E. Theissen	Sooner Or Later: Delays in Trade Reporting by Corporate Insiders
08-05	P. Linge, E. Theissen	Determinanten der Aktionärspräsenz auf Hauptversammlungen deutscher Aktiengesellschaften
08-04	N. Hautsch, D. Hess, C. Müller	Price Adjustment to News with Uncertain Precision
08-03	D. Hess, H. Huang, A. Niessen	How Do Commodity Futures Respond to Macroeconomic News?
08-02	R. Chakrabarti, W. Megginson, P. Yadav	Corporate Governance in India
08-01	C. Andres, E. Theissen	Setting a Fox to Keep the Geese - Does the Comply-or-Explain Principle Work?
2007		

07-16	M. Bär, A. Niessen, S. Ruenzi	The Impact of Work Group Diversity on Performance: Large Sample Evidence from the Mutual Fund Industry
07-15	A. Niessen, S. Ruenzi	Political Connectedness and Firm Performance: Evidence From Germany
07-14	O. Korn	Hedging Price Risk when Payment Dates are Uncertain

07-13	A.Kempf, P. Osthoff	SRI Funds: Nomen est Omen
07-12	J. Grammig, E. Theissen, O. Wuensche	Time and Price Impact of a Trade: A Structural Approach
07-11	V. Agarwal, J. R. Kale	On the Relative Performance of Multi-Strategy and Funds of Hedge Funds
07-10	M. Kasch-Haroutounian, E. Theissen	Competition Between Exchanges: Euronext versus Xetra
07-09	V. Agarwal, N. D. Daniel, N. Y. Naik	Do hedge funds manage their reported returns?
07-08	N. C. Brown, K. D. Wei, R. Wermers	Analyst Recommendations, Mutual Fund Herding, and Overreaction in Stock Prices
07-07	A. Betzer, E. Theissen	Insider Trading and Corporate Governance: The Case of Germany
07-06	V. Agarwal, L. Wang	Transaction Costs and Value Premium
07-05	J. Grammig, A. Schrimpf	Asset Pricing with a Reference Level of Consumption: New Evidence from the Cross-Section of Stock Returns
07-04	V. Agarwal, N.M. Boyson, N.Y. Naik	Hedge Funds for retail investors? An examination of hedged mutual funds
07-03	D. Hess, A. Niessen	The Early News Catches the Attention: On the Relative Price Impact of Similar Economic Indicators
07-02	A. Kempf, S. Ruenzi, T. Thiele	Employment Risk, Compensation Incentives and Managerial Risk Taking - Evidence from the Mutual Fund Industry -
07-01	M. Hagemeister, A. Kempf	CAPM und erwartete Renditen: Eine Untersuchung auf Basis der Erwartung von Marktteilnehmern

No.	Author(s)	Title
06-13	S. Čeljo-Hörhager, A. Niessen	How do Self-fulfilling Prophecies affect Financial Ratings? - An experimental study
06-12	R. Wermers, Y. Wu, J. Zechner	Portfolio Performance, Discount Dynamics, and the Turnover of Closed-End Fund Managers
06-11	U. v. Lilienfeld-Toal, S. Ruenzi	Why Managers Hold Shares of Their Firm: An Empirical Analysis
06-10	A. Kempf, P. Osthoff	The Effect of Socially Responsible Investing on Portfolio Performance
06-09	R. Wermers, T. Yao, J. Zhao	Extracting Stock Selection Information from Mutual Fund holdings: An Efficient Aggregation Approach
06-08	M. Hoffmann, B. Kempa	The Poole Analysis in the New Open Economy Macroeconomic Framework
06-07	K. Drachter, A. Kempf, M. Wagner	Decision Processes in German Mutual Fund Companies: Evidence from a Telephone Survey
06-06	J.P. Krahnen, F.A. Schmid, E. Theissen	Investment Performance and Market Share: A Study of the German Mutual Fund Industry
06-05	S. Ber, S. Ruenzi	On the Usability of Synthetic Measures of Mutual Fund Net- Flows
06-04	A. Kempf, D. Mayston	Liquidity Commonality Beyond Best Prices
06-03	O. Korn, C. Koziol	Bond Portfolio Optimization: A Risk-Return Approach

06-02	O. Scaillet, L. Barras, R. Wermers	False Discoveries in Mutual Fund Performance: Measuring Luck in Estimated Alphas
06-01	A. Niessen, S. Ruenzi	Sex Matters: Gender Differences in a Professional Setting
2005		
No.	Author(s)	Title
05-16	E. Theissen	An Analysis of Private Investors' Stock Market Return Forecasts
05-15	T. Foucault, S. Moinas, E. Theissen	Does Anonymity Matter in Electronic Limit Order Markets
05-14	R. Kosowski, A. Timmermann, R. Wermers, H. White	Can Mutual Fund "Stars" Really Pick Stocks? New Evidence from a Bootstrap Analysis
05-13	D. Avramov, R. Wermers	Investing in Mutual Funds when Returns are Predictable
05-12	K. Griese, A. Kempf	Liquiditätsdynamik am deutschen Aktienmarkt
05-11	S. Ber, A. Kempf, S. Ruenzi	Determinanten der Mittelzuflüsse bei deutschen Aktienfonds
05-10	M. Bär, A. Kempf, S. Ruenzi	Is a Team Different From the Sum of Its Parts? Evidence from Mutual Fund Managers
05-09	M. Hoffmann	Saving, Investment and the Net Foreign Asset Position
05-08	S. Ruenzi	Mutual Fund Growth in Standard and Specialist Market Segments
05-07	A. Kempf, S. Ruenzi	Status Quo Bias and the Number of Alternatives - An Empirical Illustration from the Mutual Fund Industry
05-06	J. Grammig, E. Theissen	Is Best Really Better? Internalization of Orders in an Open Limit Order Book
05-05	H. Beltran-Lopez, J. Grammig, A.J. Menkveld	Limit order books and trade informativeness
05-04	M. Hoffmann	Compensating Wages under different Exchange rate Regimes
05-03	M. Hoffmann	Fixed versus Flexible Exchange Rates: Evidence from Developing Countries
05-02	A. Kempf, C. Memmel	Estimating the Global Minimum Variance Portfolio
05-01	S. Frey, J. Grammig	Liquidity supply and adverse selection in a pure limit order book market
2004		
No.	Author(s)	Title
04-10	N. Hautsch, D. Hess	Bayesian Learning in Financial Markets – Testing for the Relevance of Information Precision in Price Discovery
04-09	A. Kempf, K. Kreuzberg	Portfolio Disclosure, Portfolio Selection and Mutual Fund Performance Evaluation
04-08	N.F. Carline, S.C. Linn, P.K. Yadav	Operating performance changes associated with corporate mergers and the role of corporate governance
04-07	J.J. Merrick, Jr., N.Y. Naik, P.K. Yadav	Strategic Trading Behaviour and Price Distortion in a Manipulated Market: Anatomy of a Squeeze
04-06	N.Y. Naik, P.K. Yadav	Trading Costs of Public Investors with Obligatory and Voluntary Market-Making: Evidence from Market Reforms

04-05	A. Kempf, S. Ruenzi	Family Matters: Rankings Within Fund Families and Fund Inflows
04-04	V. Agarwal, N.D. Daniel, N.Y. Naik	Role of Managerial Incentives and Discretion in Hedge Fund Performance
04-03	V. Agarwal, W.H. Fung, J.C. Loon, N.Y. Naik	Risk and Return in Convertible Arbitrage: Evidence from the Convertible Bond Market
04-02	A. Kempf, S. Ruenzi	Tournaments in Mutual Fund Families
04-01	I. Chowdhury, M. Hoffmann, A. Schabert	Inflation Dynamics and the Cost Channel of Monetary Transmission

centre for Financial Research

cfr/university of cologne Albertus-Magnus-Platz D-50923 cologne Fon +49[0]221-470-6995 Fax +49[0]221-470-3992 Kempf@cfr-cologne.de WWW.cfr-cologne.de